Author's response to reviews

Title: Health-related quality of life of Palestinian preschoolers in the Gaza Strip: a cross-sectional study

Authors:

Salwa G Massad (salwamassad@yahoo.com)
F. Javier Nieto (fj Nieto@wisc.edu)
Mari Palta (mpalta@wisc.edu)
Maureen Smith (maureensmith@wisc.edu)
Roseanne Clark (rclark@wisc.edu)
Abdel-Aziz M Thabet (abdelaizith@yahoo.com)

Version: 4 Date: 18 April 2011

Author’s response to reviews:

- Revised the references as indicated by the editor
- Removed track changes
- Revised the Tables that have merged cells (Tables 6 and 7)
- Checked the manuscript for any typographical errors
- The second author wants his name to appear as is: F.Javier Nieto

Thanks for considering our manuscript

Yours Truly

Salwa Massad, MCPH, PhD